Author's response to reviews

Title: High maternal mortality estimated by the sisterhood method in a rural area of Mali

Authors:

Ingvill Aa (ingvill.aa@gmail.com)
Mari A Grove (mari.aarsland@student.uib.no)
Anita H Haugsjå (ahaugsjaa@gmail.com)
Sven G. Hinderaker (Sven.Hinderaker@cih.uib.no)

Version: 5 Date: 2 August 2011

Author's response to reviews:

Dear Editor, the paper has now only a few errors as far as I can see. To follow your points:
Author list: DONE
Affiliations: DONE
Abstract: DONE
Tables: DONE. Table 2 is cited in results section, last para.
Highlighting/tracking: Please remove from the manuscript. DONE
Typography: Done

Very small modifications done in abstract (p.2), line shifts (p.7,8,9), p10 (...in contract TO...), p.12 (...more maternal deaths.)(deleter rates)

Thank you for very helpful assistance!

Regards, SGH